**Author’s response to reviews**

**Title:** Resuscitative endovascular balloon occlusion of the aorta may increase the bleeding of minor thoracic injury in severe multiple trauma patients: A case report

**Authors:**

Takaaki Maruhashi (tmaruhasi119@gmail.com)

Hiroaki Minehara (jpminet@aol.com)

Ichiro Takeuchi (itake@myad.jp)

Yuichi Kataoka (kataoka760@yahoo.co.jp)

Yasushi Asari (docya119@gmail.com)

**Version:** 2  **Date:** 09 Nov 2017

**Author’s response to reviews:**

Response to Reviewer:

We wish to express our appreciation to the Reviewer for insightful comments, which have helped us significantly improve the manuscript.

Reviewer reports:

1. At the end of the Introduction there is a need to add a paragraph that explains why this case report is presented (what is unique and adds to the medical knowledge)

Answer: Thank you for this insightful comment. According to the Reviewer’s comment, we have added the significance of submitting this case report at Introduction section of the manuscript.
2. Give complete past medical, social, family, and environmental history.

Answer: According to the Reviewer’s comment, we have added her information in detail.

3. Give detailed physical and neurological examination on admission.

Answer: According to the Reviewer’s comment, we have described her physical findings on initial arrival in detail.

4. Give all results of laboratory findings (i.e. CBC, liver and renal functions etc.)

Answer: According to the Reviewer’s comment, we have added the laboratory date on initial arrival at Table 2. With the addition of this table, the original Table 2 which explains the timeline of initial treatment has been changed to Table 3.

5. Give information about follow-up for at least 6 months.

Answer: According to the Reviewer’s comment, we have added the information of her long-term outcome at Case presentation section of the manuscript.

6. Create a separate Conclusion section at the end of the manuscript.

Answer: According to the Reviewer’s comment, we have added a separate Conclusion section.